Author’s response to reviews

Title: Safe Prescribing of Non-Steroidal Antiinflammatory Drugs in Patients with Osteoarthritis. An Expert Consensus Addressing Benefits and Gastrointestinal as well as Cardiovascular Risks

Authors:

Carmelo Scarpignato (scarpi@tin.it)
Angel Lanas (angel.lanas@gmail.com)
Corrado Blandizzi (c.blandizzi@gmail.com)
Willem F Lems (lems17@planet.nl)
Matthias Hermann (MHermann@gmx.de)
Richard H Hunt (huntr@mcmaster.ca)

Version: 2 Date: 24 January 2015

Author’s response to reviews: see over